PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Neighbourhoods for Active Kids: Study protocol for a cross-sectional
	examination of neighbourhood features and children's physical
	activity, active travel, independent mobility, and body size
AUTHORS	Oliver, Melody; McPhee, Julia; Carroll, Penelope; Ikeda, Erika;
	Mavoa, Suzanne; Mackay, Lisa; Kearns, Robin; Kyttä, Marketta;
	Asiasiga, Lanuola; Garrett, Nicholas; Lin, Judy; Mackett, Roger;
	Zinn, Caryn; Moewaka Barnes, Helen; Egli, Victoria; Prendergast,
	Kate; Witten, Karen

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.